Author's response to reviews

Title: Treatment of head lice with dimeticone 4% lotion: comparison of two formulations in a randomised controlled trial in rural Turkey

Authors:

Ozgur Kurt (oz1605@hotmail.com)
I Cuneyt Balcioglu (drcbal@yahoo.com)
Ian F Burgess (ian@insectresearch.com)
M Emin Limoncu (eminlimoncu@yahoo.com)
Nogay Girginkardesler (nogayg@yahoo.com)
Tuba Tabak (ttuba77@hotmail.com)
Hasan Muslu (caskdar@hotmail.com)
Ozge Ermis (ozgeermis@gmail.com)
M Turhan Sahin (mustafaturhan@superonline.com)
Cemal Bilac (cemalbilac@hotmail.com)
Hakan Kavur (hakankavur@yahoo.com)
Yusuf Ozbel (yusuf.ozbel@ege.edu.tr)

Version: 6 Date: 13 November 2009

Author's response to reviews:

The Tables have now been included in the main text as instructed.

The Figure has been adjusted for optimum visibility, also as per instructions.

I have made minor typographical changes to correct punctuation and spelling.